Author’s response to reviews

Title: The association between Interleukin-28B gene polymorphisms as a potential biomarker and the risk of Chronic Periodontitis in an Iranian population

Authors:

Zahra Heidari (histology_iri@yahoo.com)
Bita Moudi (bita.moudi@yahoo.com)
Hamidreza Mahmoudzadeh-Sagheb (histology@ymail.com)
Mehroosh Moudi (mehrnooshmoody@yahoo.com)

Version: 1 Date: 11 Jul 2016

Author’s response to reviews:

Approval number 6210

Declarations

Ethics approval and consent to participate: The study was approved by the Institutional Ethics Committee of the Zahedan University of Medical Sciences (approval number 6210) and written consent forms were signed by all participants.

Consent for publication: Not applicable.

Availability of data and materials: The datasets supporting the conclusions of this article are available in the Department of Histology, School of Medicine, Zahedan University of Medical Sciences, Zahedan, Iran, [http://www.zaums.ac.ir/].

Competing interests: The authors declare that they have no competing interests.

Funding: This work was supported by the vice chancellor of Research and Technology of Zahedan University of Medical Sciences (ZUMS) and Infectious Diseases and Tropical Medicine Research Center, Zahedan University of Medical Sciences [Grant No. 6210].

Authors' contribution: BM and MM carried out the molecular genetic studies. BM participated in the sequence alignment and drafted the manuscript. ZH and HM-S participated in the design of the study and performed the statistical analysis. All authors read and approved the final manuscript.

Acknowledgments: The authors would like to thank all participants who willingly participated in this study.